Author's response to reviews

Title: Anemia and hemoglobin serum levels are associated with exercise capacity and quality of life in chronic obstructive pulmonary disease

Authors:

Marcello Ferrari (marcello.ferrari@univr.it)
Lorenzo Manea (lorenzo.manea@gmail.com)
Kamel Anton (kamel.anton@gmail.com)
Paola Bruzzone (paolabruz@yahoo.it)
Mara Meneghello (mara.84@alice.it)
Francesco Zamboni (zambo987@libero.it)
Luigi Purgato (xdarkcatx@gmail.com)
Cazzoletti Lucia (lucia.cazzoletti@univr.it)
Pietro Ferrari (pietro.ferrari82@gmail.com)
Testi Renato (rtesti@qubisoft.it)

Version: 2 Date: 2 December 2014

Author's response to reviews:

Dear Reymon Ampalaya,

1) In the methods section is already reported the statement "The Institutional Review Board approved the access to patient records and patient confidentiality was maintained".

2) we added the e-mail addresses in the title page

3) we included line numbers

4) we included "competing interests" section between the conclusion and authors's contributions

5) We have no acknowledgement to include in the paper.

6) We relocated the section "Author's contributions"

Yours sincerely

Lorenzo Manea, MD